Author's response to reviews

Title: Post traumatic stress disorder and heart rate variability in Bihar flood survivors following yoga: a randomized controlled study

Authors:

Shirley Telles (shirleytelles@gmail.com)
Nilkamal Singh (naosekpamkamal@gmail.com)
Meesha Joshi (meeshie26@gmail.com)
Acharya Balkrishna (pyp@divyayoga.com)

Version: 3 Date: 2 September 2009

Author's response to reviews:

Dear Sir or Madam,

In response to the letter from the Editorial board we have made the following changes:

1. Full details of all four authors have been added

2. The trial registration number has been added to the end of the abstract. It is CTRI/2009/091/000285.

Thanking you,

Sincerely,

Shirley Telles MBBS, Ph.D.